Author’s response to reviews

Title: Relationship between breakfast consumption, BMI status and physical fitness of Ghanaian school-aged children

Authors:

Reginald Annan (reggie@imtf.org)

Solomon Sowah (sowahsolo@yahoo.com)

Charles Apprey (appreycharles@gmail.com)

Nana Ama Agyapong (nagyapong29@gmail.com)

Satoru Okonogi (Satoru.ek9@gmail.com)

Taro Yamauchi (taroy@med.hokudai.ac.jp)

Takeshi Sakurai (atsakura@mail.ecc.u-tokyo.ac.jp)

Version: 4 Date: 28 Jun 2019

Author’s response to reviews:

Response to reviewer comments has been attached as a word document. Revisions have been made to the manuscript per reviewer suggestions.